Author's response to reviews

Title: TLR3 expression correlates with apoptosis, proliferation and angiogenesis in hepatocellular carcinoma and predicts prognosis

Authors:

Ming-Ming Yuan (378842114@qq.com)
Yu-Yin Xu (xuyuyin27@aliyun.com)
Li Chen (bl1@ntu.edu.cn)
Xing-Yu Li (kawayellly@ntu.edu.cn)
Jing Qing (qj_5400@163.com)
Ying Shen (podiori@126.com)

Version: 7 Date: 18 March 2015

Author's response to reviews:

1. Please rename Introduction to Background.
Reply: I have renamed Introduction to Background.

2. Author Ying Shen is currently not included in the Authors Contribution section.
Reply: Author Ying Shen is included in the Authors Contribution section now.